Author's response to reviews

Title: Prevalence of low glomerular filtration rate, proteinuria and associated risk factors in North India using Cockcroft-Gault and Modification of Diet in Renal Disease equation: an observational, cross-sectional study

Authors:

Narinder P Singh (nanu_singh@yahoo.com)
Gopal K Ingle (jinglegk@gmail.com)
Vinay K Saini (saini@aol.in)
Ajita Jami (ajita_biochem@rediffmail.com)
Pankaj Beniwal (pankajbeniwal@yahoo.co.in)
Madan Lal (dr_madan_lal@ymail.com)
Gajender S Meena (meehags@indiatimes.com)

Version: 4 Date: 3 February 2009

Author's response to reviews: see over
To

Melissa Norton, MD
Editor-in-Chief
BMC Nephrology

February 03, 2009

Subject: Resubmission of research article manuscript: Prevalence of low glomerular filtration rate, proteinuria and associated risk factors in North India using Cockcroft-Gault and Modification of Diet in Renal Disease equation: an observational, cross-sectional study

Dear Dr. Norton

Thank you for accepting our manuscript for publication. We have made the required formatting changes and proofread the final version of manuscript also.

Sincerely
Narinder P. Singh, MD
Vinay K. Saini, MBBS